Author’s response to reviews

Title: Safety and effectiveness of a low-dose amikacin in nontuberculous mycobacterial pulmonary disease

Authors:

Maria Luisa Aznar (maznarru@gmail.com)
Marras TK (Ted.Marras@uhn.ca)
Ahmed Elshal (dr.ahmedsaeed@hotmail.com)
Mahtab Mehrabi (Matty.Mehrabi@uhn.ca)
Sarah Brode (Sarah.Brode@uhn.ca)

Version: 4 Date: 15 Apr 2019

Author’s response to reviews:

Dear Assistant Editor:

We would like to thank the comments you have made to our manuscript. We have modified the text as you suggested as follow:

Technical Comments:

1. We note that the current submission still contains some textual overlap with other previously published works, in particular:

https://doi.org/10.1080/24745332.2017.1410455

This overlap mainly exists in the Methods, Results and Discussion sections.

While we understand that this is work that you have previously published, and some of the same ideas are contained in these publications, please be aware that we cannot condone the use of text from previously published work.

Please re-phrase these sections to minimise overlap.

Methods, Results and Discussion sections have been modified to minimise overlap.